Author's response to reviews

Title: Anti-inflammatory effects of sargachromenol-rich ethanolic extract of Myagropsis myagroides on lipopolysaccharide-stimulated BV-2 cells

Authors:

Sunghee Kim (s0909bbong@nate.com)
Min-Sup Lee (kamadky@gmail.com)
Bonggi Lee (bong3257@gmail.com)
Eun-Ji Joung (eunji2007@naver.com)
Wi-Gyeong Gwon (kwk0113@naver.com)
Na-Young Yoon (dbssud@hanmail.net)
Hyeung-Rak Kim (hrkim@pknu.ac.kr)

Version: 5 Date: 20 May 2014

Author's response to reviews:

Response to editorial comment
Manuscript ID: 1286329105109763

We appreciate the editorial comments. The seaweed material was identified by seaweed taxonomist and deposited in author’s laboratory. Since we do not have a public seaweed bank, seaweed specimens were identified and deposited in author’s laboratory. Thus, we added the sentence “A voucher specimen (2012015) has been deposited in the author’s laboratory (H. R. Kim)." in the Method section (painted with yellow color). We hope that these modifications have improved the manuscript to a level of your satisfaction.